Author's response to reviews

Title: G betagamma subunits inhibit Epac-induced melanoma cell migration

Authors:

Erdene Baljinnyam (baljiner@umdnj.edu)
Masanari Umemura (umemurma@umdnj.edu)
Xie Lai-Hua (xiela@umdnj.edu)
Martha C Nowycky (nowyckmc@umdnj.edu)
Kousaku Iwatsubo (iwatsuko@umdnj.edu)

Version: 2  Date: 11 January 2011

Author's response to reviews:

We revised the manuscript according to the BMC Cancer format and included the sections on Competing interests, List of abbreviations, Acknowledgements, Authors' contributions in the main manuscript.